Author’s response to reviews

Title: Late pulmonary metastases of renal cell carcinoma immediately after post-transplantation immunosuppressive treatment: a case report

Authors:

Jose Manuel Cozar (cozarjm@supercable.es)
Natalia Aptsiauri (naptziauri@hotmail.com)
Miguel Tallada (miguel.tallada.sspa@juntadeandalucia.es)
Federico Garrido (federico.garrido.sspa@juntadeandalucia.es)
Francisco Ruiz-Cabello (fruizc@ugr.es)

Version: 8 Date: 7 April 2008

Author’s response to reviews:

Dear Editorial team,

We made the following changes to the format of our manuscript according to your recommendations:

1) each affiliation is written in full and is provided in the requested format,

2) the consent section is repositioned between the authors contributions and the acknowledgements

3) we removed figure numbers from the image files,

4) the figures are cropped to minimise white space around the image.